Author's response to reviews

Title: Perilla Extract improves GI health in a randomized placebo controlled double blind human study

Authors:

Sybille Buchwald-Werner (sybille.buchwald-werner@vitalsolutions.biz)
Hajime Fujii (hfujii@aminoup.co.jp)
Claudia Reule (c.reule@biotesys.de)
Christiane Schön (c.schoen@biotesys.de)

Version: 3 Date: 29 August 2013

Author's response to reviews:

Dear Executive Editor Mr. Foote,

We hereby re-submit a manuscript entitled “Perilla Extract improves GI health in a randomized placebo controlled double blind human study” by S. Buchwald-Werner, H. Fujii, C. Reule, C. Schön to be considered for publication as research article in BMC Complementary and Alternative Medicine (Section: Intestinal disorders) as proposed by you.

There is only 1 modification in the re-submission, which is the including of the trial registration.

Thank you for your consideration.

Sincerely,

Dr. Sybille Buchwald-Werner